Author's response to reviews

Title: A cross sectional study on Factors influencing Professionalism in nursing among nurses in Mekelle Public hospitals, Northern Ethiopia, 2012

Authors:

Atsede F Aregay (atsedefantahun@gmail.com)
Asrat D Megenta (asrat.megemta@yahoo.com)
Kahsu G Gebrekidan (kahsit12@yahoo.com)
Aynesh Z Yalew (ayuzem21@yahoo.com)
Gebre Y Abyu (gebyit45@yahoo.com)

Version: 4 Date: 10 December 2013

Author's response to reviews:

this is the study done on professionalism in nursing which is not done in Ethiopia. so publish this research helps to disseminate information easily on those factors which influence on professionalism in nursing. this contribute o have an image on the society and the profession to solve on these factors